Sleep-dependent structural neuroplasticity after a spatial navigation task: A diffusion imaging study

Thomas Villemonteix^{1,2,*}, Michele Guerreri³, Michele Deantoni^{1,4}, Evelyne Balteau⁴, Christina Schmidt⁴, Whitney Stee^{1,4}, Hui Zhang³ and Philippe Peigneux^{1,4}

¹ UR2NF-Neuropsychology and Functional Neuroimaging Research Unit affiliated at CRCN – Centre for Research in Cognition and Neurosciences and UNI - ULB Neuroscience Institute, Université Libre de Bruxelles (ULB), Brussels, Belgium

² Laboratoire Psychopathologie et Processus de Changement, EA2027, Paris 8 University, Rue de la Liberté 2, 93526 Saint-Denis, France

³Department of Computer Science & Centre for Medical Image Computing, University College London, UK3

⁴ Sleep & Chronobiology Group, GIGA-CRC-In Vivo Imaging Research Unit, University of Liège, Liège, Belgium

* Corresponding author

Abstract

Evidence for sleep-dependent changes in micro-structural neuroplasticity remains scarce, despite the fact that it is a mandatory correlate of the reorganization of learning-related functional networks. We investigated the effects of post-training sleep on structural neuroplasticity markers measuring standard diffusion tensor imaging (DTI) mean diffusivity (MD) and the revised biophysical neurite orientation dispersion and density imaging (NODDI) free water fraction (FWF) and neurite density (NDI) parameters that enable disentangling whether MD changes result from modifications in neurites or in other cellular components (e.g., glial cells). Thirty-four healthy young adults were scanned using diffusion weighted imaging [DWI] on Day1 before and after 40-minutes route learning (navigation) in a virtual environment, then were sleep deprived (SD) or slept normally (RS) for the night. After recovery sleep for 2 nights, they were scanned again (Day4) before and after 40-minutes route learning (navigation) in an extended environment. Sleep-related microstructural changes were computed on DTI (MD) and NODDI (NDI and FWF) parameters in the cortical ribbon and subcortical hippocampal and striatal regions of interest (ROIs). Results disclosed navigation learning-related decreased DWI parameters in the cortical ribbon (MD, FWF) and subcortical (MD, FWF, NDI) areas. Post-learning sleep-related changes were found at Day4 in the extended learning session (pre- to post-relearning percentage changes), suggesting a rapid sleep-related remodelling of neurites and glial cells subtending learning and memory processes in basal ganglia and hippocampal structures.

Significance statement

In this study investigating the effect of sleep on the micro-structural correlates of navigation learning, we report for the first time sleep-related effects on regional microstructural features of the brain. Our results disclose navigation learning-related decreased dMRI parameters in the cortical ribbon and subcortical areas, and post-learning sleep-related changes in the caudate nucleus and the hippocampus, which suggest a sleep-related remodelling of neurites and glial cells subtending learning and memory processes in basal ganglia and hippocampal structures.

1. Introduction

Neuroplasticity is the process by which our brain is able to structurally and functionally remodel as a result of cognitive experience (Assaf, 2018). Neuroplasticity mechanisms supporting learning and memory consolidation are operant both online, during the acquisition of novel information, and offline over two timescales after the actual learning episode (Stee & Peigneux, 2021). In the short term, a process is initiated during learning to write and stabilize memory traces within local circuits, ranging over a timescale of seconds to hours. At a longer term takes place the processes that reorganize memory traces at the systems-level, on a timescale of hours to several weeks (Wamsley, 2019). Studies from cell recording in rodents to behavioural and functional neuroimaging data in human indicate that post-learning sleep plays a key contributing role in the systems-level consolidation of recently acquired information, although offline memory consolidation can also take place during states of wakefulness (Peigneux, 2006; Rasch & Born, 2013; Wamsley, 2019). Nowadays, evidence for sleep-dependent changes in micro-structural neuroplasticity remains scarce, despite the fact that it is a mandatory correlate of the reorganization of functional learning-related networks (Stee & Peigneux, 2021).

Among key cognitive systems, spatial memory is responsible for recording information about one's environment and spatial orientation (Do et al., 2021). Functional brain imaging studies typically found spatial navigation associated with activity in a distributed cerebral network including the hippocampus, dorsal striatum and entorhinal, parahippocampal, retrosplenial and frontal cortices (Epstein et al., 2017). Hippocampus and entorhinal cortex are proposed to subtend the creation of a maplike spatial code, i.e., a representation of the structure of the environment. Parahippocampal and retrosplenial cortices are thought to provide critical inputs anchoring the cognitive map to fixed environmental landmarks (Epstein et al., 2017; Gahnstrom & Spiers, 2020). Spatial coding is not limited to medial temporal structures; spatial representations are also associated with orbito-frontal (Wikenheiser et al., 2021) and/or parietal (Brodt et al., 2018; Schott et al., 2019) cortex activity. Besides map-based representations, spatial navigation can also rely on stimulus–response associations supporting the development of habit-based route strategies. Human functional Magnetic Resonance Imaging (fMRI) and animal lesion studies evidenced a role for the dorsal striatum in the formation of habits and habit-based navigation (Orban et al., 2006; Rauchs, Orban, Balteau, et al., 2008; Wunderlich et al., 2012). The dorsal striatum plays a complementary role during flexible navigation, through the implementation of reinforcement learning computations supporting encoding of transitions structure and forward planning in the caudate nuclei (Gahnstrom & Spiers, 2020).

The two-stage theory of systems-level memory consolidation proposes that declarative and spatial memory traces are initially encoded into the hippocampus, which serves as a transient storage site, and are then gradually transferred to the neocortex for long-term storage (Born & Wilhelm, 2012). Sleep is thought to play a key role in this process, thanks to the oscillatory rhythms occurring during deep non-rapid eye movement (NREM) sleep (Binder et al., 2019; Born & Wilhelm, 2012; Buzsáki, 1998). Rodent studies showed the spontaneous replay of learning-related neural patterns of hippocampal place cell activity (e.g., Ólafsdóttir et al., 2018). Accordingly, in human, increased activity was found during NREM sleep in navigation-related hippocampal and parahippocampal regions in participants trained to explore a virtual city before sleep, as compared to subjects trained on another, nonhippocampus-dependent procedural memory task (Peigneux et al., 2004). Besides, fMRI and magnetoencephalography (MEG) studies provided evidence that post-learning sleep leads to a reorganization of the brain patterns underlying the retrieval of topographical or declarative memories (Orban et al., 2006; Rauchs, Orban, Schmidt, et al., 2008; Urbain et al., 2016). Indeed, functional MRI studies showed that navigation in a recently learned environment, initially subtended by hippocampusbased spatial strategies, becomes more contingent on striatal activity a few days later when participants are allowed to sleep on the first night after learning (Orban et al., 2006; Rauchs, Orban, Schmidt, et al., 2008), suggesting that sleep favoured a shift toward a habit-based, cognitively less demanding navigation strategy.

Memory-related neuroplasticity mechanisms have been mostly investigated in humans measuring functional brain responses, but the underlying micro-structural brain changes are more elusive to detect at limited time scales such as hours or even days. In this respect, advances in diffusion magnetic resonance imaging (dMRI) techniques have opened new avenues to explore the (micro)structural neuroplasticity changes supporting memory processes in the human brain (Assaf, 2018). Diffusion tensor imaging (DTI) studies showed that mean diffusivity (MD), a global marker of tissue microstructure, can track the microstructural changes associated with learning in the short term. Investigating spatial learning-related neuroplasticity, Sagi et al. (2012) evidenced decreased mean diffusivity (MD) in the grey matter bilaterally in the hippocampus and parahippocampus in participants repeatedly trained on the same route for 120 minutes in a video game task, as compared to control participants driving random routes A similar study evidenced decreased MD in hippocampus, precuneus, superior frontal gyrus, left insula, supramarginal, superior temporal and fusiform gyri following a 60-minutes training session in which participants driving 20 different routes (Keller & Just, 2016). To the best of our knowledge, only one study tested an effect of sleep on micro-structural plasticity; results showed decreased mean cortical MD after 12 hours of intense training on a driving simulation game and executive function tasks, an effect magnified after 24 h of task practice combined with sleep deprivation, that reverted after recovery sleep (Bernardi et al., 2016).

An inherent limitation of DTI is that even if sensitive to the tissue's microstructure, it merely provides a composite view of all cellular components within the measured voxel and does not provide information on its internal partitions. An alternative model to conventional DTI approach is the neurite orientation dispersion and density imaging (NODDI) model, a biophysical model specifically designed to describe both WM and GM which provides unique information regarding the extent of neurite orientation variability in the space (Zhang et al., 2012). As mentioned above, evidence for sleep-dependent changes in micro-structural neuroplasticity remains scarce (Stee & Peigneux, 2021). In the present dMRI study, we investigated the effects of a post-training sleep manipulation on structural markers of neuroplasticity. Participants were scanned using diffusion weighted imaging (DWI) on Day 1 before and after a 40-minutes route learning (navigation) session in a virtual environment (Peigneux et al., 2004, 2006), then sleep deprived (SD) for the whole night or allowed to sleep normally (RS). They all had then 2 nights with normal sleep. On Day 4, they were scanned again using DWI before and after a 40-minutes route learning (navigation) session in an extended version of the virtual environment. We used dMRI derived metrics from both DTI and NODDI approaches to assess microstructural changes in

the cortical ribbon and in subcortical regions of interest (ROIs). Results disclosed learning-related decreased parameters in the cortical ribbon and subcortical areas, and post-learning sleep-related changes in the extended learning session at Day 4 in ROIs caudate nucleus and hippocampus, suggesting an early sleep-related remodelling of both neurites and glial processes in subcortical regions.

2. Material and Methods

2.1. Participants and Procedure

34 healthy volunteers gave written informed consent to participate in this study approved by the Ethics Committee of the University of Liege, Belgium (reference B70720156274). They were paid for their participation. All procedures were performed following relevant guidelines and regulations. One participant was excluded due to abnormal structural MRI scan, and one participant due to a corrupted data file in one of the diffusion imaging acquisitions. The final sample consisted of 32 right-handed (Laterality Quotient Score > 70 (Oldfield, 1971) participants (17 males; mean age, 23.5 years; range, 19–31 years). Self-reported questionnaires assessed sleep habits (PSQI (Buysse et al., 1989)) and circadian topology (MEQ (Horne & Ostberg, 1976)) for the past month before inclusion. Exclusion criteria were extreme chronotype (MEQ score > 69 or < 31) and bad sleep quality (PSQI score > 5), prior history of neurological or psychiatric disorder, or consumption of psychotropic drugs.



Figure 1. Experimental Protocol. On Day 1, participants are scanned with diffusion-weighted imaging before (DWI-1) and after (DWI-2) out-of-scanner learning of Level 1 of the virtual town (left part of the figure). They are also tested for navigation performance in a fMRI session (fMRI-1) at the end of learning before DWI-2. Participants then either sleep normally (RS) or are sleep deprived (SD) during post-learning Night 1, and then sleep normally on Nights 2 and 3. On Day 4, RS and SD participants are scanned again with diffusion (DWI-3) then tested again for navigation performance in a fMRI session (fMRI-2). Afterward, they are trained on an extended version (both Level 1 and 2; right part of the figure) of the virtual town and then scanned a last time (DWI-4). Finally, participants are tested on the extended version of the virtual town outside of the MRI environment (Retrieval extended, not illustrated). Top left and right panels provide an aerial representation of the town map (not seen by participants who navigated from within the environment, see sample pictures) with Level 1 (left, day 1) and 2 (right, day 4) environments. For more details on the environment maps illustrations from the participants' perspective within the task, see the Supplementary Material section. Also, extended description and detailed illustrations of the navigation task and behavioural testing can be found in Peigneux et al. (2004), Orban et al. (2006), Rauchs et al. (2008) and Deantoni et al. (2021).

The experimental timeline is illustrated in Figure 1. On Day 1, sessions were conducted in the afternoon from 13:00 to 20:00. On that day, mean wake-up time as assessed by self-reported questionnaires was 08:43 (SD = 44 minutes). Participants were first scanned with a diffusion-weighted imaging session (DWI-1), then had four successive 8-minutes blocks outside of the scanner to explore and learn the Level 1 of a virtual city (see below). They were then scanned with functional magnetic resonance imaging (fMRI) while performing a navigation test where they had to reach as fast as possible 10 targets within the learned environment (Day 1 Navigation Performance Score), and then scanned again in a second diffusion imaging session (DWI-2). Participants were then randomly assigned either to the regular sleep (RS; n = 15) or to the sleep deprivation (SD; n = 17) condition. Groups did not differ on age (RS: mean(m) = 22.27, standard-deviation(sd) = 1.83; SD: m = 22.34, sd = 2.87; p = .921) or sex

ratio (RS: 8 males ; SD: 9 males ; $\chi^2 = .982$, p = .630). Randomisation was achieved using a random number generation algorithm in matlab, generating a series of 0 (RS) and 1 (SD). Participants in the RS group were allowed to sleep as usual at home and requested to follow regular sleep habits for 3 experimental nights, based on their usual schedule. In the SD group, participants were kept awake in the laboratory during the first post-learning night. During this SD night, their physical activity was maintained as low as possible under constant supervision by the experimenters. Participants remained most of the time in a seated position, and were allowed to read, chat, play quiet games and watch movies. They received isocaloric meals at regular intervals, and water ad libitum. At 8:00 am, they were allowed to leave the lab, and were instructed to have non-dangerous daytime activities and abstain from napping until bedtime. They then slept normally at home and were requested to follow regular sleep habits the two following nights. Sleep quality and quantity for the nights from before the first learning session (Day 1) to the last testing session (Day 4) was assessed using a daily standardised questionnaire (Ellis et al., 1981), controlled by visual inspection of continuous actimetric recordings (ActiGraph, wGT3X-BT Monitor, USA). On Day 4, SD and RS participants came back to the laboratory for a second experimental session in the afternoon, at the same time than at Day 1. First, they were scanned with diffusion imaging (DWI-3). Afterwards, they were scanned with functional magnetic resonance imaging (fMRI) while performing a navigation test where they had to reach as fast as possible 10 targets within the learned L1 environment (Day 4 Navigation Performance Score). They were then trained outside of the MRI scanner for an additional four 8-minutes blocks on an extended version of the navigation task (Level 1 plus Level 2 of the town; Figure 1), in which the targets and environments of the initial training were incorporated. They were then scanned again with the diffusion sequence (DWI-4) and finally tested on the extended version of the navigation task outside of the MRI environment (Day 4 Relearning score). A high-resolution structural MRI scan was also acquired for each participant at the end of the brain imaging session (Day 1 or Day 4). In this protocol, 5-minutes long resting-state fMRI sessions were also acquired after each DWI session. Resting-state fMRI results have been reported in a previous publication (Deantoni et al., 2021) and will not be discussed here.

2.2. Navigation task

The navigation task is described in detail elsewhere (Deantoni et al., 2021), only essential information is reported here. Participants were trained in a previously used virtual navigation environment (Orban et al., 2006; Peigneux et al., 2006; Rauchs, Orban, Balteau, et al., 2008), that features a complex town made of two levels communicating through two teleports (Figure 1). Each level is composed of three districts, distinct from each other both visually (architecture, objects ...) and auditorily (background sounds and music) and containing each one target location. Participants are explicitly instructed to learn their way in the city by moving freely within the environment. At both Day 1 and Day 4 learning sessions, participants are exposed to a test trial at the end of each 8-minute exploration block, i.e., they are assigned a starting point and instructed to reach a given target location as fast as possible in a maximal time of 28 s (Day 1) or 60 s (Day 4), i.e., the shortest possible duration from starting point to target destination within L1 (Day 1) or from L1 to L2 (Day 4). At the end of the four 8-minute blocks of free exploration, participants are exposed to 10 test trials in a row to estimate their navigation performance. A quantitative performance measure is computed for each trial as the shortest distance remaining between the subject's actual location at the end of the trial time limit and the location of the target destination, based on video screen recordings of their navigation (Deantoni et al., 2021; Orban et al., 2006; Peigneux et al., 2006; Rauchs, Orban, Balteau, et al., 2008). Participants were also tested on 10 L1 navigation trials during a fMRI session on Day 1 after learning before DWI-2 and on Day 4 before relearning after DWI-3.

2.3. Diffusion Imaging

2.3.1. Data Acquisition

Brain MRI data were acquired on a whole-body 3T scanner (Magnetom Prisma, Siemens Medical Solutions, Erlangen, Germany). Diffusion MRI images were acquired with a 64-channel head coil, using a multislice 2D single-shot spin-echo echo-planar imaging with monopolar gradients. Parallel imaging

with the iPAT technique was used with a reduction factor of 2, one excitation, and 2mm interleaved axial slices with no gap at an isotropic resolution of 2x2 mm on a 96x96 matrix. The echo time (TE) and repetition time (TR) were 69ms and 7400ms. There were 70 slices (b-values: 650, 1000 and 2000 sec/mm2 with 15, 30 and 60 directions respectively; 13 additional interleaved brain volumes were also acquired with no diffusion weighting, and two extra b=0 volumes were acquired with opposite phase encoding for susceptibility induced distortion correction). For anatomical reference, a high-resolution T1-weighted structural image was acquired for each subject (T1-weighted 3D magnetization-prepared rapid gradient echo (MPRAGE) sequence, TR = 1900 ms, TE = 2.19 ms, FA = 9 deg, inversion time (TI) = 900 ms, FoV = 256 × 240 mm2, matrix size = 256 × 240 × 224, voxel size = 1 × 1 × 1 mm3, acceleration factor in phase-encoding direction R = 2). Processing and analyses of task-based fMRI data are reported in Deantoni et al. (2021).

2.3.2. MRI data processing

In this section we detail the steps towards the estimation of the diffusion MRI (dMRI) derived metrics as well as towards the determination of correspondence across subjects and across timepoints. Microstructural changes associated with fast learning episodes have been previously reported in subcortical as well as in neocortical areas (Brodt et al., 2018; Keller & Just, 2016; Sagi et al., 2012), hence our analysis included both. We derived two streams of analysis optimised for each of them. Microstructural changes in the cortex were estimated via a surface-based analysis by projecting the dMRI metrics on the cortical surface. Microstructural changes in the grey matter of the hippocampus, caudate nucleus and putamen were assessed via a region of interest (ROI)-based analysis. In both cases we leveraged the anatomical information from T1-weighted (T1-w) images to establish inter-subject correspondence. Intra-subject group correspondence between different timepoints was established through DWI images using the full tensor information (Zhang et al. 2007).

2.3.2.1 Pre-processing

Correspondence across subjects, for both the surfaced-based and the ROI-based analysis, was determined via the anatomical (T1w)images FreeSurfer software via (https://surfer.nmr.mgh.harvard.edu/, v7.1.1). To improve the anatomical image analysis robustness, pre-processing steps inspired by the minimal pre-processing pipeline for the Human Connectome Project (Glasser et al., 2013) were applied prior to the FreeSurfer analysis. First, anatomical images were reoriented to match the orientation of the standard MNI152 template images. Next, they were cropped to a smaller FOV removing the neck and the lower part of the head, and finally, they were aligned to match the MNI152 template via the anterior to posterior commissure ("ACPC") alignment procedure (Glasser et al., 2013). These steps were implemented via the FMRIB Software Library (FSL, v6.0) (Jenkinson et al., 2012; Smith et al., 2004; Woolrich et al., 2009). A preliminary analysis showed that the anatomical image marked bias field affected the FreeSurfer analysis outcome quality, hence all images underwent an extra bias field correction step via the N4ITK algorithm (Tustison et al., 2010). Finally, in order to assist the final registrations to MNI space and to assist FreeSurfer with its brain mask creation process, a rough brain mask was generated using the method described in Glasser et al (2013).

For dMRI, the pre-processing steps included susceptibility induced distortion correction, eddy current induced distortion correction and head motion correction. Susceptibility induced distortions were estimated from b=0 volumes with opposed phase encoding via FSL's topup (Andersson et al., 2003). FSL's eddy routine was used to simultaneously correct for susceptibility and eddy current induced distortions, as well as for head motion (Andersson & Sotiropoulos, 2016).

2.3.2.2 diffusion MRI (dMRI) models

As a reminder, NODDI models brain microarchitecture using three complementary components within a voxel volume: (1) intra-neurite (modelling the space bounded by the neurite membranes and myelin sheaths), (2) extra-neurite (modelling the space outside of neurites, including glial cells and somas), and (3) free water (accounting for the presence of cerebrospinal fluid or oedema in the tissues) (Zhang et al., 2012). NODDI's output is a set of indices aimed at describing neurite morphology. The neurite density index (NDI) accounts for the intra-neurite compartment signal fraction. The free-water fraction (FWF) is the free-water compartment signal fraction. The orientation dispersion index (ODI) measures the directional variability of neurites. In this work we used a revised version of NODDI which addresses some of the original technique limitations when extended to GM (Guerreri et al., 2018, 2020). NODDI revised version has been shown to be more sensitive to subtle microstructural changes in the cortex, compared to its original version (Guerreri et al., 2021).

Diffusion tensor imaging (DTI) is another widespread MRI technique that has been used to assess neural plasticity in the past (Basser & Pierpaoli, 1996). Unlike NODDI, DTI considers each voxel as a single compartment, giving only a composite picture of the tissues. The compartment is modelled by a tensor that describes the 3D profile of diffusion. DTI derived metrics are the mean diffusivity (MD), which quantifies the average diffusivity (the "size" of the tensor) in each voxel, and the fractional anisotropy (FA), quantifying the overall amount of directional preference of diffusion in each voxel (the "shape" of the tensor).

2.3.2.3 Model fitting

The NODDI model was fitted to all available diffusion data (b = 0, 650, 1000, 2000 s/mm2) via a modified version of the NODDI-toolbox (http://mig.cs.ucl.ac.uk/index.php?n=Tutorial.NODDImatlab) implemented in Matlab (The Math Works, Inc. MATLAB. Version 2019b). For a robust parameter estimation (Jelescu et al., 2016; Zhang et al., 2012), the intra-neurite bulk diffusivity was fixed to the value of 2.28 μ m2/ms, which is supported by previous studies using a non-conventional, generalised version of dMRI acquisitions (Guerreri et al., 2020). DTI was fitted to the data with b-values up to 1000 s/mm2 via FSL's DTIFIT tool.

2.3.2.4 Intra-subject group alignment and diffusion to anatomical space registration

To define intra-subject correspondence across DW images from different timepoints, we implemented a robust approach to minimise longitudinal image processing biases such as interpolation asymmetry (Yushkevich et al., 2010). In brief, the approach finds a subject-specific target space that represents the centre of all the time points. The spatial normalisation exploits the full tensor information (Zhang et al. 2007). This approach has the advantage of treating all time points equally, which is an essential requirement in longitudinal image processing (Reuter et al., 2012). The implementation is available at https://github.com/CIG-UCL/dtitk/tree/master/addons/IntraSubjGrpAlign.

Next, for each subject, we defined the correspondence transformation from each of the DWI time point and the corresponding T1w image available. This is done in three two steps: first, we determined a subject-specific DW template into the time point target space defined previously, which was representative of all time points; second, we registered the template to the T1w image via an affine transformation; third, we found the transformation that maps each DWI into the corresponding anatomical image by composing the two transformations.

Finally, all the dMRI derived metrics were resampled from each individual time point space into the corresponding anatomical space in one single step to minimise interpolation effects.

2.3.2.5 Inter-subject group correspondence

FreeSurfer software was used to establish the correspondence between subjects by segmenting subcortical areas (Fischl et al., 2002, 2004) as well as registering each subject cortical surface to a template surface (Dale et al., 1999; Fischl & Dale, 2000). The pre-processed anatomical images underwent the FreeSurfer routine. A visual quality assessment was performed at the end of each of the three main sections of the routine. Ad hoc interventions were carried out when necessary to ensure good quality of the segmentations and of the surface placement. The hippocampus is known to have an important role in spatial navigation (Epstein et al., 2017). In particular, different sub-structures are believed to have different functions (Harland et al., 2017). To account for this, we used FreeSurfer segmentHA_T1 routine (Iglesias et al., 2015) to further divide the segmented hippocampi into head, body and tail.

We assessed sub-cortical GM plasticity-dependent microstructural changes by comparing the dMRI-derived metric average values in each of the considered ROIs across time points and subjects. For each subject, the mean values were computed into each time point native diffusion space to minimise any possible resampling bias. The FreeSurfer derived segmentations were resampled into each time

point space inverting the transformation used to bring the dMRI derived maps into the anatomical space. Based on prior navigation studies reviewed above (Epstein et al., 2017), our subcortical ROI analysis focused on five bilateral regions, namely left and right head, body, and tail of the hippocampus, putamen and caudate.

In the cortical GM, we assessed plasticity-dependent microstructural changes vertex-wise via a surface-based analysis. To increase the sensitivity to subtle changes, it is common practice to perform spatial smoothing which increases the statistical power (Brodt et al., 2018; Tavor et al., 2020). In a surface-based analysis framework the smoothing step occurs on the surface. Hence, compared to conventional voxel-wise volumetric approaches, this procedure has two main advantages: first, it preserves cortical folding spatial continuity i.e., it prevents mixing of information from cortical regions which are spatially close but topologically distant; second, it prevents mixing of information from non-cortical regions such as WM and CSF. In order to implement the surface-based analysis we projected each of the dMRI derived metrics which were resampled into the anatomical space onto the cortical surface. Specifically, to reduce partial volume contamination (Fukutomi et al., 2018), the mid-thickness surface was used. The mid-thickness surface was obtained by expanding the pial surface i.e., the surface marking the boundary between white matter (WM) and grey matter (GM), midway through the cortical thickness via FreeSurfer's "mris_expand" routine. Finally, the projected maps were resampled into the template surface space. Smoothing with a 6mm FWHM Gaussian filter was performed to all the surface maps in this space before statistical analysis.

2.3.3. Statistical analyses (dMRI data)

All analyses were conducted in the general linear model (GLM) framework for each of the three diffusion imaging parameters of interest (i.e., mean diffusivity [MD], neurite density index [NDI], free water fraction [FWF]). First, we investigated learning-related short-term changes from before (DWI-1) to after (DWI-2) the Day 1 Learning session in the whole sample (N = 32). Second, we investigated potential post-training sleep effects running analysis with between-subject factor Group (RS vs. SD)

and within-subject factor either Pre-Learning sessions, i.e., first DWI scan at Day 1 (DWI-1) and Day 4 (DWI-3), or Relearning-related Session (DWI-3 vs. DWI-4).

In the cortical ribbon, contrasts were computed using FreeSurfer's (http://surfer.nmr.mgh.harvard.edu/) surface-based stream applying a vertex-based approach. Cluster-wise correction for multiple comparisons (p<.05) was performed running a non-parametric permutation simulation using FreeSurfer's *mri_glmft-sim* command with 10,000 simulations and a vertex wise cluster-forming threshold of p<.001. Peak coordinates of significant clusters are reported based on the MNI system. For the sake of completeness, clusters associated with learning at the uncorrected level (p<.001) are reported as Supplementary Material.

Based on the FreeSurfer volume-based subcortical segmentation stream, we also examined the above mentioned contrasts on MD, NDI and FWF values in our five subcortical left and right regions-of-interest (ROIs) identified as key actors of the navigation learning network (Epstein et al., 2017): the head, body and tail of the hippocampus, the caudate nucleus and the putamen. For each indice, paired sample t-tests were conducted to examine potential learning effects (DWI-1 vs. DWI-2) and extended learning effects (DWI-3 vs. DWI-4). A repeated measure ANOVA was computed with group (SD vs. RS) as between-subject factor and within-subject factor either Pre-Learning sessions, i.e., first DWI scan at Day 1 (DWI-1) and Day 4 (DWI-3), or Relearning-related Session (DWI-3 vs. DWI-4) to investigate potential post-training sleep effects. Results on a priori ROIs are reported at p-level < .05, uncorrected.

3. Results

3.1. Behavioural Results

3.1.1. Sleep Data

Sleep duration and quality were estimated by means of self-reports of the nights preceding Day 1 and Day 4. Visual inspection of actimetric data of Day 2 was conducted to detect potential naps in the sleep deprived groups. No such event was detected. Mean sleep duration was not significantly different

between RS and SD groups on the night before Day 1 (RS = 439 ± 63 min, SD = 453 ± 41 min; p = 0.252; df = 31) and the night before Day 4 (RS = 437 ± 55 min, SD = 458 ± 44 min; p = 0.23). Within each group, duration of each night did not differ significantly (RS, p = 0.825; SD, p = 0.744, ; df = 31). Likewise, a similar analysis conducted on subjective sleep quality assessed by means of a 6-point scale failed to evidence between-group differences. Subjective sleep quality was equivalent between groups for each night (night before Day 1: RS = 3.7 ± 1.2 , SD = 3.5 ± 1.1 ; p = 0.549; df = 31; night before Day4: RS = 3.6 ± 0.7 , SD = 3.7 ± 0.9 ; p = 0.921; ; df = 31). Finally, within-group comparisons did not reveal any significant difference in sleep quality between both nights (RS p = 0.82; SD p = 0.155). These results indicate that sleep patterns observed the night before the SD episode were restored before being scanned again in the delayed session.

3.1.2. Navigation Performance

As a reminder, for each of the 10 place finding tests administered at Day 1 (Immediate retrieval) and Day 4 (Delayed retrieval), participants were instructed to reach, as fast as possible, a given Level 1 target from one designated starting point within 28 s. Navigation performance was estimated as the distance remaining to reach destination at the end of the trial. A repeated measure ANOVA computed on individual mean scores with group (SD vs. RS) as between-subject factor and retrieval session (Immediate vs. Delayed) as within-subject factor did not reveal any significant main or interaction effect (mean distance score at Day 1: SD = 17.08 arbitrary units (standard deviation 8.23) vs RS = 15.19 (13.32); df = 31 ; Day 4 SD = 17.59 (11.2) vs. RS = 15.32 (12.1); df = 31, main effect of time : p = .884; groupXday interaction effect: p = .643). Likewise, at testing after the second learning session on the extended version of the town (distance to destination after the 60 seconds deadline), mean performance did not statistically differ (t = -0.2; p = 0.837; ; df = 31) between RS = 15.7 (19.20) and SD = 17.2 (19.6) conditions.

3.3. Diffusion Weighted Imaging (DWI) results

3.3.1. Learning-related structural changes (Day 1; DWI-1 vs. DWI-2)

3.3.1.1. Cortical Ribbon

Learning on Day 1 was associated with significantly decreased MD and FWF in the cortical ribbon (Figure 2 [top panel], Supplementary Table 1.a & 1.b). FWF significantly decreased bilaterally in the occipital pole, middle occipital gyrus, angular gyrus, central sulcus expanding in the precentral gyrus in the left hemisphere, and in the superior part of the precentral sulcus. In the right hemisphere, FWF decreased in the parieto-occipital sulcus expanding in the cuneus and precuneus, the posterior transverse collateral sulcus, the lateral occipitotemporal sulcus, in three clusters in the superior temporal sulcus, in the posterior dorsal cingulate gyrus and in the middle frontal gyrus. In the left hemisphere, clusters were located in the middle occipital sulcus, occipito-superior and transversal sulci, in the lingual and fusiform gyri, the intraparietal sulcus, the precuneus, the posterior part of the lateral fissure, the superior part of the precentral and the pars marginalis of the cingulate sulcus. MD similarly decreased in the same locations than FWF, to the exception of one cluster in the left superior temporal sulcus. No significant changes were found for NDI.



Figure 2: Learning- and Extended Learning-related structural changes (cortical ribbon). Significant clusters of decreases in MD and FWF after Learning on Day 1 (upper panel) and Extended Learning on Day 4 (lower panel) in the cortical ribbon after cluster-wise correction for multiple comparisons. Significance levels are on a -log(p) scale. See Supplemental Figure S1 for results without cluster-wise correction for multiple comparison on MD, FWF and NDI parameters.

3.3.1.2. Subcortical ROIs

In subcortical ROIs, significant changes were found in the right Caudate, in which FWF decreased by 1.9% (DWI-1 mean(SD) = .1284(.0243) vs. DWI-2 mean(SD) = .1260(.0238), t = 2.09, p = .044, df = 31, g = .1) and NDI decreased by 0.8% (DWI-1 = .3453(.0329) vs. DWI-2 = .3426(.0321), t = 2.3, p = .029, df = 31, g = .08), and in the left and right Body of the Hippocampus, in which FWF decreased by

1.9% and 1.7% (Left: DWI-1 = .1672(.0282) vs. DWI-2 = .1641(.0281), t = 2.38, p = .023, df = 31, g = .11; Right: DWI-1 = .1839(.0220) vs. DWI-2 = .1807(.0243), t = 2.08, p = .045, df = 31, g = .13).

3.3.2. Sleep-related effects in Pre-learning vs. Pre-relearning (DWI-1 vs. DWI-3 X RS vs. SD)

No significant effect was found in the cortical ribbon for any of the three indices. A main effect of sleep was detected in the subcortical ROIs, where a significant change was found in the right Putamen; NDI decreased by 0.8% and 0.7% (Noddi1, mean(SD): .3942(.0270), Noddi3, mean(SD): .3915(.0285), t = 2.63, p = .013, df = 31, g = .09). No significant interaction effects were found either in the cortical ribbon or in the subcortical ROIs, for any of the three indices.

3.3.3. Sleep-related effects on Extended Learning (DWI-3 vs. DWI-4 X RS vs. SD)

3.3.3.1. Cortical Ribbon

The analysis evidenced a main session effect (DWI-3 vs DWI-4), but no Group (RS vs. SD) nor Group X Session interaction effects in the cortical ribbon. Regarding the main session effect, extended learning on Day 4 was associated with significantly decreased MD and FWF in the cortical ribbon (Figure 2 [bottom panel], Supplementary Table 2.a & 2.b). FWF decreased bilaterally in the occipito-superior and transversal sulcus, the parieto-occipital sulcus, the superior temporal and the central sulcus. In the right hemisphere, clusters were localised in the lingual gyrus, the cuneus gyrus, the middle occipital gyrus, the subparietal sulcus, the precentral sulcus and the circular sulcus. In the left hemisphere, clusters were found in the occipital pole, the calcarine sulcus, the middle occipital sulcus and gyrus, the superior occipital gyrus, the subparietal sulcus, the intraparietal and transversal sulcus and gyrus, the superior occipital gyrus, the subparietal sulcus, the right left hemisphere, clusters were found in the occipital pole, the calcarine sulcus, the middle occipital sulcus and gyrus, the superior occipital gyrus, the subparietal sulcus, the intraparietal and transversal sulcus and gyrus, the superior occipital gyrus, the subparietal sulcus, the intraparietal and transversal sulcus and gyrus, the superior occipital gyrus, the subparietal sulcus, the intraparietal and transversal sulcus and gyrus, the posterior part of the lateral fissure and in the pars marginalis of the cingulate sulcus. Clusters found significant for FWF encompassed all clusters significant for MD to the exception of one cluster in the central sulcus (x = -14.5; y = -29.1; z = 58.9).

3.3.3.2. Subcortical ROIs

The analysis evidenced main Session (DWI-3 vs DWI-4) and Group by Session interaction effects, but no main Group (RS vs. SD) effect. Group by Session interaction effects on FWF were found in the right Caudate (RS 2.7% decrease = -.0033 (.0050) vs. SD 0.9% increase = +.0011 (.0062); t = 2.26, p = .031, df = 30), and on NDI in the right Caudate (RS 0.2% decrease = -.0007(.0057) vs. SD 1.3% increase = +.0045 (.0062); t = 2.46, p = .020, df = 30) and the left Head of the Hippocampus (RS 1.3% decrease= -.0029 (.0076) vs. SD 1.8% increase = .0039 (.0087); t = 2.35, p = .025, df = 30) (Figure 3).



Figure 3. Sleep-related effects on Extended Learning (DWI-3 vs DWI-4 vs RS vs SD). Box & violin plots showing sleep-related changes in the Regular Sleep (RS) vs. Sleep Deprived (SD) group in percentage for a) Free Water Fraction (FWF) in the Right Caudate (RS 2.7% decrease vs. SD 0.9% increase), b) Neurite Density Index (NDI) in the Right Caudate (RS 0.2% decrease vs. SD 1.3% increase) and c) NDI in the Left Hippocampal Head (RS 1.3% decrease vs. SD 1.8% increase).

Regarding the main Session effect (i.e., relearning-related changes), MD decreased over time by 0.4%in the right Caudate (DWI-3 mean(SD) = .0007296(.00001351) vs. DWI-4 mean(SD) = .0007265(.00001221), t = 2.78, p = .009, df = 31, g = .23), by 0.5\% in the Left and Right Body of the Hippocampus (Left Hpc DWI-3 = .0007882(.00001780) vs. DWI-4 = .0007843(.00001713), t = 2.09, p = .044, df = 31; g = .22; Right Hpc DWI-3 = .0008015(.00001913) vs. DWI-4 = .0007975(.00001805), t = 2.14, p = .040, df = 31, g = .21) and by 0.4% in the right Hippocampus Head (DWI-3 = .0008048(.00001206) vs. DWI-4 = .0008012(.00001094), t = 2.67, p = .012, df = 31, g = 0.3). FWF decreased by 1.5% in the left Hippocampus Head (DWI-3 = .1647(.0243) vs. DWI-4 = .1622(.0230), t = 2.19, p = .036, df = 31, g = .1), by 2.3% and 2.5% in the Left and Right Body of the Hippocampus, respectively (Left Body DWI-3 = .1685(.0298) vs. DWI-4 = .1647(.0298), t = 2.85, p = .008, df = 31, g = .12; Right Body DWI-3 = .1851(.0243) vs. DWI-4 = .1805(.0254), t = 3.31, p = .002, df = 31, g = .18), and by 2.6% and 3.1% in the Left Right Tail and of the Hippocampus (Left Tail DWI-3 = .1755(.0411) vs. DWI-4 = .1709(.0405), t = 2.77, p = .009, df = 31, g = .11; Right Tail DWI-3 = .1835(.0413) vs. DWI-4 = .1779(.0427), t = 3.99, p < .001, df = 31, g = .13). Finally, NDI increased by 0.6% in the Left Putamen (DWI-3 = .3954(.0246) vs. DWI-4 = .3979(.0245), t = -2.19, p = .036, df = 31, g = .11).

Discussion

In the present study, we used advanced dMRI analysis techniques to investigate the offline microstructural changes associated with learning and extended learning following sleep in a spatial navigation task.

In line with previous studies (e.g., Orban et al., 2006; Rauchs, Orban, Schmidt, et al., 2008), behavioural performance was similar between the sleep deprived (SD) and regular sleep (RS) group, both after initial learning, and when tested after exploring an extended version of the town. Our behavioural results suggest that (1) both groups gained knowledge of the virtual town that persisted 3 days after learning, that (2) SD on the first post-training night did not overtly alter subjects' ability to find their way in the town at delayed retrieval (which does not preclude the use of qualitatively different neural and cognitive strategies for a similar performance), and that (3) learning novel routes and integration with prior knowledge in the extended version of the town was quantitatively similar in the RS and SD conditions.

In the whole sample, learning and extended learning episodes were associated with an overall pattern of MD (DTI) and FWF (NODDI) decreases in the cortical ribbon. Maps of MD and FWF changes were highly coherent (see Supplementary figure 1). As a reminder, the DTI mean diffusivity (MD)

parameter quantifies the average diffusivity in each voxel. MD was proposed in previous studies to be a relevant and sensitive marker of tissue microstructure changes associated with learning. However, the biological interpretation of MD changes is not straightforward. Indeed, the calculated tensor is an average of all cellular components within the measured voxel, which can be affected by several parameters of tissue density or tissue organisation, such as synaptogenesis, changes in the morphometry of axons, dendrites and glial processes, or alterations in cell body size and shape (Sagi et al., 2012). In a previous study, Tavor et al. (2013) evidenced short-term plasticity changes associated with spatial navigation learning using both a conventional DTI approach and the biophysical composite hindered and restricted diffusion (CHARMED) approach, that models the tissues in a voxel as the integration of multiple components (i.e., hindered and restricted compartments) and provides an estimate for each of them (Sagi et al., 2012). In their study, decreased MD was found again after 120 minutes car-racing in the left hippocampus, bilateral parahippocampal gyrus and bilateral insula. MD changes were paralleled by an increased water volume in the restricted compartment (Fr) of the CHARMED model, tentatively proposed to reflect the remodelling of dendrite and glial processes in the grey matter (GM) (Tavor et al., 2013). Furthermore, increased Fr was detected in hippocampal and parahippocampal areas already after 45 minutes of learning, while DTI failed to reflect such changes, suggesting CHARMED a more sensitive method to study structural plasticity. However, CHARMED was mostly designed for white matter (WM) modelling and is thus less suited for describing GM changes.

The NODDI model assumes that water molecules in neuronal tissue can be considered within three separate compartments: (a) free water, representing CSF; (b) restricted water representing neurites; and (c) hindered water, representing diffusion within glial cells, neuronal cell bodies and the extracellular environment. This enables estimation of parameters including the free water fraction index (FWF), the neurite density index (NDI) and the orientation dispersion index (ODI). In the NODDI model, FWF is the component that accounts for the presence of cerebrospinal fluid or oedema in the tissues volume, besides the intra- and extra-neurite spaces (Zhang et al., 2012). Hence, in the present study, reduced FWF thus theoretically entails more space occupied by the two other compartments. our results suggests that structural modifications took place in the extra-neurite space including glial cells, in line with the interpretation proposed in Tavor et al. (2013) based on the CHARMED framework. Astrocytes are key elements of synaptic function, and exhibit rapid changes following synaptic activity (Theodosis et al., 2008). Evidence of structural plasticity in astrocyte morphology was previously found in rats trained on the Morris water maze (Blumenfeld-Katzir et al., 2011). Rapid structural modifications are also possible in dendritic spines, as formation of dendritic spines after training was shown to occur within the hour (Xu et al., 2009).

Both navigation learning and relearning episodes evoked overall similar patterns of microstructural changes, that are in line with fMRI findings in spatial navigation studies (Epstein et al. 2017). On the medial surface, clusters were mostly localised in the occipital and parietal lobes, while clusters found on the lateral surface were mostly localised in the occipital, parietal, temporal and posterior frontal areas (see Supplementary figure 1). In subcortical ROIs, learning and relearning were associated with a set of changes in MD and FWF again, but also in NDI. In the hippocampus, MD decreases were found for the extended learning episode only, whereas FWF decreased both in learning and extended (re)learning episodes. These findings concur with previous studies that associated spatial navigation learning with MD decreases in hippocampal areas (Keller & Just, 2016; Sagi et al., 2012; Tavor et al., 2013). A discrepant finding is the absence of decreased MD in the first learning episode. Besides possible statistical power effects, it should be acknowledged that the spatial navigation task we used was different from previous studies using dMRI to explore the microstructural changes associated with spatial learning. Indeed, it is likely more complex as participants have to learn to find their way in a maze, as opposed to learning to repeatedly drive a single road. As for other subcortical ROIs, FWF reductions were found in the right caudate after the learning episode, and MD reductions after the extended learning episode. Finally, NDI decreased in the right caudate after learning, but increased in the left putamen after relearning. Increased NDI at relearning may reflect a plasticity-related restructuration of the intra-neurite space bounded by neurite membranes and myelin sheaths, consistent with concomitantly reduced FWF.

Regarding sleep-related consolidation effects, no differential changes were found in the cortical ribbon at extended learning when comparing the RS and SD conditions. Besides statistical power issues, this null finding may indicate that sleep-dependent structural reorganisation at the cortical level was not yet implemented and/or detectable up to 3 days after the sleep manipulation, in line with fMRI findings that hippocampo-neocortical transfer of information was not observed two days after sleep deprivation but only weeks to month later in young adults (Gais et al., 2007). At variance, in subcortical ROIs, microstructural changes associated with extended learning were modulated by post-learning sleep. In the left hippocampus head, NDI decreased in participants allowed to sleep after learning (RS) but increased in the sleep-deprived ones (SD). Also in the right caudate nucleus, NDI increased in the SD condition. Finally, FWF decreased in the right Caudate in the RS condition and increased in the SD condition. Hence, our data suggest that sleep following learning influences the microstructural underpinning of later extended learning episodes in which novel information is integrated into existing knowledge, in such a way that opposite patterns can be detected in key areas supporting spatial navigation learning. Decreased hippocampal NDI paralleled by stabilised NDI (and reduced FWF) in the caudate in the RS condition may to some extent reflect the gradual shift from a hippocampusdependent spatial strategy to a more integrated and automated striatal response-based strategy, as found in fMRI studies (Orban et al., 2006; Rauchs, Orban, Schmidt, et al., 2008).

One potential limitation of the present study is the lack of a control condition investigating the microstructural changes associated with exposure to spatial navigation without learning. Including such a control condition may have helped identifying clusters of changes playing a key role in learning and extended learning, as opposed to brain areas involved in spatial navigation. Without such a control group, it must be acknowledged that some of the detected changes could be caused by scan repetition and unspecific practice effects between scans, besides actual changes associated with learning, although prior studies evidenced satisfying reliability in repeated DWI measurements (see e.g., Sagi et al., 2012). Another potential limitation is the fact that our ROI analysis did not correct for the examination of multiple regions of interest, increasing the risks of type 1 error despite the fact that we worked on a priori defined ROIs. For these reasons, we consider our ROI-based results as preliminary. On the other

hand, the investigation of learning and extended learning changes in the cortical ribbon was corrected for multiple comparisons. Also on the positive side, our processing pipeline was optimised separately for cortical and sub-cortical GM analysis. Previous studies (e.g., Brodt et al., 2018; Sagi et al., 2012; Tavor et al., 2013, 2020) used voxel-based non-linear registration to define correspondence across subjects, combined with spatial smoothing to increase statistical power. We believe that such an approach may not be optimal for at least two reasons. First, volumetric smoothing entails the risk of mixing information from non-GM regions such as CSF and WM. Second, in the highly convoluted cortical foldings, volumetric smoothing can induce mixing of information from cortical regions which are spatially close but topologically distant. In the present study, we used two specific analysis streams for cortical and subcortical GM analyses, aimed at avoiding such limitations. In the surface-based analysis in particular, the smoothing step occurs on the surface after the projection of the relevant information from the cortical GM voxels is performed, hence preventing information mixing as described above. For its part, ROI analysis does not involve any smoothing step and thus should not be affected by the reported information mixing issue.

To sum up, our results disclose navigation learning-related decreased dMRI parameters in the cortical ribbon and subcortical areas. Post-learning sleep-related changes were found in the extended learning session in the caudate nucleus and the hippocampus, suggesting a sleep-related remodelling of neurites and glial cells subtending learning and memory processes in basal ganglia and hippocampal structures, that may precede further reorganisation at the cortical level.

Conflict of Interest Statement

This study was supported by the Fonds de la Recherche Scientifique (F.R.S.-F.N.R.S.) and the Fonds Wetenschappelijk Onderzoek – Vlaanderen (F.W.O.) under the Excellence of Science (EOS) Project (MEMODYN, No. 30446199) coordinated by P.P.. W.S. is supported by an F.N.R.S. Aspirant Research Fellowship. M.G. received postdoctoral salary support from the EOS MEMODYN project. At the time of the study, T.V. was funded by an Université Libre de Bruxelles (ULB) Individual Fellowship. M.D.

is supported by the Fonds pour la Recherche dans l'Industrie et l'Agriculture (FRIA) Fellowship. C.S. is FNRS Research Associate. The authors have no conflict of interest to declare.

Author contributions

Thomas Villemonteix: Conceptualization, Formal analysis, Investigation, Writing - Original Draft, Writing - Review & Editing, Visualization, Project administration ; Michele Guerreri: Methodology, Software, Formal analysis, Writing - Original Draft, Writing - Review & Editing ; Michele Deantoni: Formal analysis, Investigation, Writing - Original Draft, Writing - Review & Editing, Visualization, Project administration ; Evelyne Balteau: Methodology ; Software ; Project administration ; Resources ; Writing - Original Draft ; Christina Schmidt: Methodology ; Software ; Project administration ; Resources ; Writing - Original Draft, Supervision ; Whitney Stee: Formal analysis ; Writing - Original Draft ; Writing - Review & Editing ; Visualization ; Project administration ; Hui Zhang: Methodology, Software, Writing - Original Draft, Writing - Review & Editing, Supervision, Project administration, Funding acquisition ; Philippe Peigneux: Conceptualization, Methodology, Software, Formal analysis, Investigation, Writing - Original Draft, Writing - Review & Editing, Visualization, Supervision, Project administration, Funding acquisition.

Data accessibility statement

The data presented in this study are available on request from the corresponding author. The data are not yet publicly available due to scientific agreement between parties who process related elements in the dataset.

References

Andersson, J. L. R., Skare, S., & Ashburner, J. (2003). How to correct susceptibility distortions in spin-echo echo-planar images: Application to diffusion tensor imaging. *NeuroImage*, 20(2), 870–888. https://doi.org/10.1016/S1053-8119(03)00336-7

Andersson, J. L. R., & Sotiropoulos, S. N. (2016). An integrated approach to correction for offresonance effects and subject movement in diffusion MR imaging. *NeuroImage*, *125*, 1063–1078. https://doi.org/10.1016/j.neuroimage.2015.10.019 Assaf, Y. (2018). New dimensions for brain mapping. *Science (New York, N.Y.)*, 362(6418), 994–995. https://doi.org/10.1126/science.aav7357

Basser, P. J., & Pierpaoli, C. (1996). Microstructural and physiological features of tissues elucidated by quantitative-diffusion-tensor MRI. *Journal of Magnetic Resonance. Series B*, *111*(3), 209–219. https://doi.org/10.1006/jmrb.1996.0086

Bernardi, G., Cecchetti, L., Siclari, F., Buchmann, A., Yu, X., Handjaras, G., Bellesi, M., Ricciardi, E., Kecskemeti, S. R., Riedner, B. A., Alexander, A. L., Benca, R. M., Ghilardi, M. F., Pietrini, P., Cirelli, C., & Tononi, G. (2016). Sleep reverts changes in human gray and white matter caused by wake-dependent training. *NeuroImage*, *129*, 367–377. https://doi.org/10.1016/j.neuroimage.2016.01.020

Binder, S., Mölle, M., Lippert, M., Bruder, R., Aksamaz, S., Ohl, F., Wiegert, J. S., & Marshall, L. (2019). Monosynaptic Hippocampal-Prefrontal Projections Contribute to Spatial Memory Consolidation in Mice. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*, *39*(35), 6978–6991. https://doi.org/10.1523/JNEUROSCI.2158-18.2019

Blumenfeld-Katzir, T., Pasternak, O., Dagan, M., & Assaf, Y. (2011). Diffusion MRI of structural brain plasticity induced by a learning and memory task. *PloS One*, *6*(6), e20678. https://doi.org/10.1371/journal.pone.0020678

Born, J., & Wilhelm, I. (2012). System consolidation of memory during sleep. *Psychological Research*, *76*(2), 192–203. https://doi.org/10.1007/s00426-011-0335-6

Brodt, S., Gais, S., Beck, J., Erb, M., Scheffler, K., & Schönauer, M. (2018). Fast track to the neocortex: A memory engram in the posterior parietal cortex. *Science (New York, N.Y.)*, *362*(6418), 1045–1048. https://doi.org/10.1126/science.aau2528

Buysse, D. J., Reynolds, C. F., Monk, T. H., Berman, S. R., & Kupfer, D. J. (1989). The Pittsburgh Sleep Quality Index: A new instrument for psychiatric practice and research. *Psychiatry Research*, 28(2), 193–213. https://doi.org/10.1016/0165-1781(89)90047-4

Buzsáki, G. (1998). Memory consolidation during sleep: A neurophysiological perspective. *Journal of Sleep Research*, 7 *Suppl 1*, 17–23. https://doi.org/10.1046/j.1365-2869.7.s1.3.x

Dale, A. M., Fischl, B., & Sereno, M. I. (1999). Cortical surface-based analysis. I. Segmentation and surface reconstruction. *NeuroImage*, *9*(2), 179–194. https://doi.org/10.1006/nimg.1998.0395

Deantoni, M., Villemonteix, T., Balteau, E., Schmidt, C., & Peigneux, P. (2021). Post-Training Sleep Modulates Topographical Relearning-Dependent Resting State Activity. *Brain Sciences*, *11*(4). https://doi.org/10.3390/brainsci11040476

Do, T.-T. N., Lin, C.-T., & Gramann, K. (2021). Human brain dynamics in active spatial navigation. *Scientific Reports*, *11*(1), 13036. https://doi.org/10.1038/s41598-021-92246-4

Ellis, B. W., Johns, M. W., Lancaster, R., Raptopoulos, P., Angelopoulos, N., & Priest, R. G. (1981). The St. Mary's Hospital sleep questionnaire: A study of reliability. *Sleep*, *4*(1), 93–97. https://doi.org/10.1093/sleep/4.1.93 Epstein, R. A., Patai, E. Z., Julian, J. B., & Spiers, H. J. (2017). The cognitive map in humans: Spatial navigation and beyond. *Nature Neuroscience*, 20(11), 1504–1513. https://doi.org/10.1038/nn.4656

Fischl, B., & Dale, A. M. (2000). Measuring the thickness of the human cerebral cortex from magnetic resonance images. *Proceedings of the National Academy of Sciences*, *97*(20), 11050–11055. https://doi.org/10.1073/pnas.200033797

Fischl, B., Salat, D. H., Busa, E., Albert, M., Dieterich, M., Haselgrove, C., van der Kouwe, A., Killiany, R., Kennedy, D., Klaveness, S., Montillo, A., Makris, N., Rosen, B., & Dale, A. M. (2002). Whole brain segmentation: Automated labeling of neuroanatomical structures in the human brain. *Neuron*, *33*(3), 341–355. https://doi.org/10.1016/s0896-6273(02)00569-x

Fischl, B., Salat, D. H., van der Kouwe, A. J. W., Makris, N., Ségonne, F., Quinn, B. T., & Dale, A. M. (2004). Sequence-independent segmentation of magnetic resonance images. *NeuroImage*, *23 Suppl 1*, S69-84. https://doi.org/10.1016/j.neuroimage.2004.07.016

Fukutomi, H., Glasser, M. F., Zhang, H., Autio, J. A., Coalson, T. S., Okada, T., Togashi, K., Van Essen, D. C., & Hayashi, T. (2018). Neurite imaging reveals microstructural variations in human cerebral cortical gray matter. *NeuroImage*, *182*, 488–499. https://doi.org/10.1016/j.neuroimage.2018.02.017

Gahnstrom, C., & Spiers, H. (2020). Striatal and hippocampal contributions to flexible navigation in rats and humans. *Brain and Neuroscience Advances*, *4*, 239821282097977. https://doi.org/10.1177/2398212820979772

Gais, S., Albouy, G., Boly, M., Dang-Vu, T. T., Darsaud, A., Desseilles, M., Rauchs, G., Schabus, M., Sterpenich, V., Vandewalle, G., Maquet, P., & Peigneux, P. (2007). Sleep transforms the cerebral trace of declarative memories. *Proceedings of the National Academy of Sciences*, *104*(47), 18778–18783. https://doi.org/10.1073/pnas.0705454104

Glasser, M. F., Sotiropoulos, S. N., Wilson, J. A., Coalson, T. S., Fischl, B., Andersson, J. L., Xu, J., Jbabdi, S., Webster, M., Polimeni, J. R., Van Essen, D. C., Jenkinson, M., & WU-Minn HCP Consortium. (2013). The minimal preprocessing pipelines for the Human Connectome Project. *NeuroImage*, *80*, 105–124. https://doi.org/10.1016/j.neuroimage.2013.04.127

Guerreri, M, Szczepankiewicz, F, Lampinen, B, Nilsson, M, Palombo, M, Capuani, S, & Zhang, H. (2018). *Revised NODDI model for diffusion MRI data with multiple b-tensor encodings, International Society for Magnetic Resonance in Medicine*. https://discovery.ucl.ac.uk/id/eprint/10082897/

Guerreri, M, Szczepankiewicz, F, Lampinen, B, Nilsson, M, Palombo, M, & Zhang, H. (2020). Tortuosity assumption not the cause of NODDI's incompatibility with tensor-valued diffusion encoding, International Society for Magnetic Resonance in Medicine. Sydney, Australia. https://discovery.ucl.ac.uk/id/eprint/10082897/

Guerreri, M, Villemonteix, T, Stee, W, Balteau, E, Peigneux, P, & Zhang, H. (2021). Quantifying tissue microstructural changes associated with short-term learning using model-based *diffusion MRI. International Society for Magnetic Resonance in Medicine.* https://discovery.ucl.ac.uk/id/eprint/10082897/

Harland, B., Contreras, M., & Fellous, J.-M. (2017). A Role for the Longitudinal Axis of the Hippocampus in Multiscale Representations of Large and Complex Spatial Environments and Mnemonic Hierarchies. In *The Hippocampus—Plasticity and Functions*. IntechOpen. https://doi.org/10.5772/intechopen.71165

Horne, J. A., & Ostberg, O. (1976). A self-assessment questionnaire to determine morningnesseveningness in human circadian rhythms. *International Journal of Chronobiology*, 4(2), 97–110.

Iglesias, J. E., Augustinack, J. C., Nguyen, K., Player, C. M., Player, A., Wright, M., Roy, N., Frosch, M. P., McKee, A. C., Wald, L. L., Fischl, B., Van Leemput, K., & Alzheimer's Disease Neuroimaging Initiative. (2015). A computational atlas of the hippocampal formation using ex vivo, ultra-high resolution MRI: Application to adaptive segmentation of in vivo MRI. *NeuroImage*, *115*, 117–137. https://doi.org/10.1016/j.neuroimage.2015.04.042

Jelescu, I. O., Veraart, J., Fieremans, E., & Novikov, D. S. (2016). Degeneracy in model parameter estimation for multi-compartmental diffusion in neuronal tissue. *NMR in Biomedicine*, *29*(1), 33–47. https://doi.org/10.1002/nbm.3450

Jenkinson, M., Beckmann, C. F., Behrens, T. E. J., Woolrich, M. W., & Smith, S. M. (2012). FSL. *NeuroImage*, 62(2), 782–790. https://doi.org/10.1016/j.neuroimage.2011.09.015

Keller, T. A., & Just, M. A. (2016). Structural and functional neuroplasticity in human learning of spatial routes. *NeuroImage*, *125*, 256–266. https://doi.org/10.1016/j.neuroimage.2015.10.015

Long, X., & Zhang, S.-J. (2021). A novel somatosensory spatial navigation system outside the hippocampal formation. *Cell Research*, *31*(6), 649–663. https://doi.org/10.1038/s41422-020-00448-8

Ólafsdóttir, H. F., Bush, D., & Barry, C. (2018). The Role of Hippocampal Replay in Memory and Planning. *Current Biology*, 28(1), R37–R50. https://doi.org/10.1016/j.cub.2017.10.073

Oldfield, R. C. (1971). The assessment and analysis of handedness: The Edinburgh inventory. *Neuropsychologia*, *9*(1), 97–113. https://doi.org/10.1016/0028-3932(71)90067-4

Orban, P., Rauchs, G., Balteau, E., Degueldre, C., Luxen, A., Maquet, P., & Peigneux, P. (2006). Sleep after spatial learning promotes covert reorganization of brain activity. *Proceedings of the National Academy of Sciences of the United States of America*, 103(18), 7124–7129. https://doi.org/10.1073/pnas.0510198103

Peigneux, P. (2015). Neuroimaging studies of sleep and memory in humans. *Current Topics in Behavioral Neurosciences*, 25, 239–268. https://doi.org/10.1007/7854_2014_326

Peigneux, P., Laureys, S., Fuchs, S., Collette, F., Perrin, F., Reggers, J., Phillips, C., Degueldre, C., Del Fiore, G., Aerts, J., Luxen, A., & Maquet, P. (2004). Are spatial memories strengthened in the human hippocampus during slow wave sleep? *Neuron*, 44(3), 535–545. https://doi.org/10.1016/j.neuron.2004.10.007 Peigneux, P., Orban, P., Balteau, E., Degueldre, C., Luxen, A., Laureys, S., & Maquet, P. (2006). Offline persistence of memory-related cerebral activity during active wakefulness. *PLoS Biology*, *4*(4), e100. https://doi.org/10.1371/journal.pbio.0040100

Rasch, B., & Born, J. (2013). About sleep's role in memory. *Physiological Reviews*, 93(2), 681–766. https://doi.org/10.1152/physrev.00032.2012

Rauchs, G., Orban, P., Balteau, E., Schmidt, C., Degueldre, C., Luxen, A., Maquet, P., & Peigneux, P. (2008). Partially segregated neural networks for spatial and contextual memory in virtual navigation. *Hippocampus*, *18*(5), 503–518. https://doi.org/10.1002/hipo.20411

Rauchs, G., Orban, P., Schmidt, C., Albouy, G., Balteau, E., Degueldre, C., Schnackers, C., Sterpenich, V., Tinguely, G., Luxen, A., Maquet, P., & Peigneux, P. (2008). Sleep modulates the neural substrates of both spatial and contextual memory consolidation. *PloS One*, *3*(8), e2949. https://doi.org/10.1371/journal.pone.0002949

Reuter, M., Schmansky, N. J., Rosas, H. D., & Fischl, B. (2012). Within-subject template estimation for unbiased longitudinal image analysis. *NeuroImage*, *61*(4), 1402–1418. https://doi.org/10.1016/j.neuroimage.2012.02.084

Sagi, Y., Tavor, I., Hofstetter, S., Tzur-Moryosef, S., Blumenfeld-Katzir, T., & Assaf, Y. (2012). Learning in the fast lane: New insights into neuroplasticity. *Neuron*, 73(6), 1195–1203. https://doi.org/10.1016/j.neuron.2012.01.025

Schott, B. H., Wüstenberg, T., Lücke, E., Pohl, I.-M., Richter, A., Seidenbecher, C. I., Pollmann, S., Kizilirmak, J. M., & Richardson-Klavehn, A. (2019). Gradual acquisition of visuospatial associative memory representations via the dorsal precuneus. *Human Brain Mapping*, *40*(5), 1554–1570. https://doi.org/10.1002/hbm.24467

Smith, S. M., Jenkinson, M., Woolrich, M. W., Beckmann, C. F., Behrens, T. E. J., Johansen-Berg, H., Bannister, P. R., De Luca, M., Drobnjak, I., Flitney, D. E., Niazy, R. K., Saunders, J., Vickers, J., Zhang, Y., De Stefano, N., Brady, J. M., & Matthews, P. M. (2004). Advances in functional and structural MR image analysis and implementation as FSL. *NeuroImage*, *23 Suppl 1*, S208-219. https://doi.org/10.1016/j.neuroimage.2004.07.051

Stee, W., & Peigneux, P. (2021). Post-learning micro- and macro-structural neuroplasticity changes with time and sleep. *Biochemical Pharmacology*, *191*, 114369. https://doi.org/10.1016/j.bcp.2020.114369

Tavor, I., Botvinik-Nezer, R., Bernstein-Eliav, M., Tsarfaty, G., & Assaf, Y. (2020). Short-term plasticity following motor sequence learning revealed by diffusion magnetic resonance imaging. *Human Brain Mapping*, *41*(2), 442–452. https://doi.org/10.1002/hbm.24814

Tavor, I., Hofstetter, S., & Assaf, Y. (2013). Micro-structural assessment of short term plasticity dynamics. *NeuroImage*, *81*, 1–7. https://doi.org/10.1016/j.neuroimage.2013.05.050

Tustison, N. J., Avants, B. B., Cook, P. A., Zheng, Y., Egan, A., Yushkevich, P. A., & Gee, J. C. (2010). N4ITK: Improved N3 bias correction. *IEEE Transactions on Medical Imaging*, *29*(6), 1310–1320. https://doi.org/10.1109/TMI.2010.2046908

Urbain, C., De Tiège, X., Op De Beeck, M., Bourguignon, M., Wens, V., Verheulpen, D., Van Bogaert, P., & Peigneux, P. (2016). Sleep in children triggers rapid reorganization of memory-related brain processes. *NeuroImage*, *134*, 213–222. https://doi.org/10.1016/j.neuroimage.2016.03.055

Wamsley, E. J. (2019). Memory Consolidation during Waking Rest. *Trends in Cognitive Sciences*, 23(3), 171–173. https://doi.org/10.1016/j.tics.2018.12.007

Wikenheiser, A. M., Gardner, M. P. H., Mueller, L. E., & Schoenbaum, G. (2021). Spatial Representations in Rat Orbitofrontal Cortex. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*, *41*(32), 6933–6945. https://doi.org/10.1523/JNEUROSCI.0830-21.2021

Woolrich, M. W., Jbabdi, S., Patenaude, B., Chappell, M., Makni, S., Behrens, T., Beckmann, C., Jenkinson, M., & Smith, S. M. (2009). Bayesian analysis of neuroimaging data in FSL. *NeuroImage*, *45*(1 Suppl), S173-186. https://doi.org/10.1016/j.neuroimage.2008.10.055

Wunderlich, K., Dayan, P., & Dolan, R. J. (2012). Mapping value based planning and extensively trained choice in the human brain. *Nature Neuroscience*, *15*(5), 786–791. https://doi.org/10.1038/nn.3068

Xu, T., Yu, X., Perlik, A. J., Tobin, W. F., Zweig, J. A., Tennant, K., Jones, T., & Zuo, Y. (2009). Rapid formation and selective stabilization of synapses for enduring motor memories. *Nature*, 462(7275), 915–919. https://doi.org/10.1038/nature08389

Yushkevich, P. A., Avants, B. B., Das, S. R., Pluta, J., Altinay, M., Craige, C., & Alzheimer's Disease Neuroimaging Initiative. (2010). Bias in estimation of hippocampal atrophy using deformationbased morphometry arises from asymmetric global normalization: An illustration in ADNI 3 T MRI data. *NeuroImage*, *50*(2), 434–445. https://doi.org/10.1016/j.neuroimage.2009.12.007

Zhang, H., Avants, B.B., Yushkevich, P.A., Woo, J.H., Wang, S., McCluskey, L.F., Elman, L.B., Melhem, ER, Gee JC. High-dimensional spatial normalization of diffusion tensor images improves the detection of white matter differences: an example study using amyotrophic lateral sclerosis. IEEE Trans Med Imaging. 2007 Nov;26(11):1585-97. doi: 10.1109/TMI.2007.906784

Zhang, H., Schneider, T., Wheeler-Kingshott, C. A., & Alexander, D. C. (2012). NODDI: Practical in vivo neurite orientation dispersion and density imaging of the human brain. *NeuroImage*, *61*(4), 1000–1016. <u>https://doi.org/10.1016/j.neuroimage.2012.03.072</u>